Reviewer #1:

**Scientific Quality:** Grade B (Very good)

**Language Quality:** Grade A (Priority publishing)

**Conclusion:** Minor revision

**Specific Comments to Authors:** This is an interesting study of plasma brain natriuretic peptide and platelet parameters in COPD. The manuscript is very well written. Please take attention about the abbreviations, and some minor language polishings. Thank you.

Thank you very much for your suggestions. We have made modifications.

Reviewer #2:

**Scientific Quality:** Grade C (Good)

**Language Quality:** Grade B (Minor language polishing)

**Conclusion:** Minor revision

**Specific Comments to Authors:** Platelets are known to contribute to thrombus formation, and pulmonary artery thrombosis is the main factor that leads to death in patients with chronic obstructive pulmonary disease complicated with pulmonary heart disease. For properly guiding clinical treatment, it is crucial to screen appropriate biochemical indices for patients diagnosed to have combined chronic obstructive pulmonary disease and pulmonary heart disease. In this study, the authors explored the effect of plasma BNP and platelet parameters on assessing the health condition of patients with chronic obstructive pulmonary disease and pulmonary heart disease. This design of this study is good. The results are very interesting and well discussed. A minor revision is required. Comments: 1. There are some minor language polishing, which should be revised. 2. The title are too long, please revise it, it can be changed to “Plasma brain natriuretic peptide and platelet parameters in chronic obstructive pulmonary disease and pulmonary heart disease patients”. 3. Please add a short background to the abstract. 4. Please move the tables to the end of the manuscript. 5. Please discuss the limit of this study.

Thank you very much for your suggestions. We have made modifications.

1. We've polished the language

2. We have revised the title

3. We've added background to the abstract

4. We have moved the tables to the end of the article

5. We added a discussion of the limitations of the article